Author's response to reviews

Title: The readiness of the national health laboratory system in supporting care and treatment of HIV/AIDS in Tanzania

Authors:

Leonard E.G. Mboera (lmboera@nimr.or.tz)
Deus R.S. Ishengoma (deusishe@yahoo.com)
Andrew M. Kilale (kilale@hotmail.com)
Isolide S. Massawe (isolidesylvester@yahoo.com)
Acleus S.M. Rutta (malinzi55@yahoo.com)
Gibson B. Kagaruki (gkagaruki@gmail.com)
Erasmus Kamugisha (erasmuskamugisha@yahoo.com)
Vito Baraka (vitobaraka@gmail.com)
Celine I. Mandara (drceiline2003@yahoo.com)
Godlisten S. Materu (materu.godlisten@yahoo.com)
Stephen M. Magesa (smagesa@hotmail.com)

Version: 3
Date: 16 January 2015

Author's response to reviews:

We have include the following:
1. A separate section of Conclusion
2. Conflict of Interest
3. Authors’ contribution
4. Acknowledgement